# Bank efficiency in the digital age: The role of financial technology in Tanzanian banks

Omary Juma Ally <sup>1,\*</sup> <sup>(D)</sup>, Yusuph Kulindwa <sup>2</sup> <sup>(D)</sup>, and Lucas Mataba <sup>3</sup> <sup>(D)</sup>

- 1 Bank of Tanzania Academy, Mwanza, Tanzania, email: darvinommy@gmail.com
  - 2 Moshi Co-operative University, Department of Economics and Statistics, Kilimanjaro, Tanzania, email: yusuphkulindwaj@yahoo.com
- 3 Moshi Co-operative University, Department of Banking and Finance, Kilimanjaro, Tanzania, email: matabalucasi@gmail.com
- \* Correspondence at: Omary Juma, Department of Academic Research and Consultancy, Bank of Tanzania Academy, Mwanza, Tanzania P.O.BOX 131, Mwanza City, Tanzania, email: darvinommy@gmail.com

Abstract: The global rise of financial technology offers opportunities and challenges for banking businesses, including Tanzanian banks. This study examines the influence of a bank's FinTech index on the efficiency of 30 Tanzanian commercial banks categorized as large, medium, and small from 2010–2021. Using panel data and a two-step Generalized Method of Moments (GMM) estimator, the study finds that the FinTech index measuring banks' financial technology development significantly enhances efficiency across all banks, with the largest impact on large banks due to their high financial technology development. However, medium and small banks face challenges in financial technology development, resulting in a negative relationship between the FinTech index and the efficiency of banks. The study emphasizes the need for regulatory frameworks supporting financial technology integration in the core banking systems, especially for smaller and medium banks. It highlights the importance of collaboration and risk management to enhance bank efficiency and financial stability.

Keywords: financial technology, efficiency, Tanzanian commercial banks, FinTech index.

# 1. Introduction

Globally, financial technology (FinTech) has transformed the financial industry, significantly altering how financial services are delivered (Arnaut & Bećirović, 2023). For instance, in 2022, global investments in FinTech reached \$164 billion, driving innovations such as artificial intelligence-driven risk management and blockchain technology (KPMG,2023). In the banking sector, this integration is evident in adopting digital lending platforms, mobile banking, and automated credit scoring systems that have enhanced customer experience and operational efficiency. For example, 70% of banks worldwide now offer digital-only services, reflecting the rapid integration of FinTech solutions into traditional banking practices (Deloitte, 2023). Despite this evident growth, empirical evidence on how FinTech influences the efficiency of commercial banks across different categories, such as large, medium, and small banks, remains limited (Al-Shari & Lokhande, 2023; Elsaid, 2023).

Bank efficiency has become an increasingly important factor in the global financial landscape, influencing the sustainability and competitiveness of banking institutions (Broby, 2021). Efficient banks are better positioned to allocate resources, minimize operational costs, and offer innovative services, contributing to long-term stability (Badunenko *et al.*, 2021; Lee *et al.*, 2021). For instance, Badunenko et al. (2021) highlighted that efficient resource allocation boosts bank profitability and mitigates systemic risks and insights critical to understanding efficiency measures in this study. Similarly, Lee et al. (2021) analyzed cost-reduction strategies in banks of varying sizes,

Check for updates

**Citation:** Juma Ally, O., Kulindwa, Y., & Mataba, L. (2025). Bank efficiency in the digital age: The role of financial technology in Tanzanian banks. *Modern Finance*, 3(1), 1-24.

Accepting Editor: Adam Zaremba

Received: 4 December 2024 Accepted: 22 January 2025 Published: 23 January 2025



**Copyright:** © 2025 by the authors. This article is an open-access article distributed under the terms and conditions of the Creative Commons Attribution 4.0 International (CC BY 4.0) license (https://creativecommons .org/licenses/by/4.0/). providing a framework for this study's evaluation of efficiency differences across bank categories. In a globally interconnected banking system, inefficiencies can escalate quickly and have far-reaching consequences for different bank sizes (Abdu, 2022; Habib *et al.*, 2020). The role of FinTech in enhancing bank efficiency is particularly significant as it streamlines operations, improves resource allocation, and reduces manual errors through automation (Sajid *et al.*, 2023; Dwivedi *et al.*, 2021).

In Africa, the banking sector increasingly recognizes FinTech's transformative potential, driving innovations in financial services. In South Africa, banks have adopted technologies such as mobile banking, blockchain for secure transactions, artificial intelligence (AI) for customer service and fraud detection, and application program interfaces (APIs) for seamless integration (Kshetri, 2021). These technologies streamline operations, reduce transaction time, and enhance customer satisfaction. Similarly, in East Africa, Tanzania and Kenya have leveraged FinTech to revolutionize banking with mobile money platforms, such as M-Pesa, promoting financial inclusion among unbanked populations. Banks have also embraced cloud computing for efficient data management, biometric authentication for security, and data analytics to optimize decision-making and tailor financial products (Ntwiga, 2020). However, limited empirical research exists on how FinTech integration within a bank influences the efficiency of Tanzanian banks across different categories, highlighting the need to address this gap in enhancing banking performance.

In sub-Saharan Africa, FinTech can potentially drive substantial progress in financial services, particularly by improving efficiency and extending access to previously underserved markets (IMF, 2019). FinTech can make banks more efficient, improve service quality, and expand their customer base (Murinde *et al.*, 2022). This is critical for banks to continue playing a crucial role in their countries' economic development. FinTech promises to enhance bank efficiency by bridging information gaps, reducing costs through automation, and reaching unbanked customers (Chan *et al.*, 2022; Murinde *et al.*, 2022). Conversely, inefficient banks can hinder economic growth. When banks mismanage resources or operate inefficiently, they slow down the flow of capital and limit growth (Isik & Uygur, 2021). This highlights the importance of understanding how the FinTech development of commercial banks can enhance efficiency, optimize resource allocation, and ensure the banking sector's contribution to overall economic well-being.

According to Statista (2024), Tanzania's Fintech sector is witnessing significant growth, particularly in digital financial services. The digital assets market is projected to be the largest segment, with assets under management (AUM) expected to reach USD 5.70 million in 2024. The average AUM per user in the Digital Investment market for the same year is estimated at USD 79.35, indicating an increasing adoption of digital investment solutions. Furthermore, the Digital Assets market is anticipated to grow at a revenue rate of 23.90% by 2025, showing its rapid development. In the digital payment segment, user numbers are projected to rise significantly, reaching 11.60 million by 2028. These trends underscore the expanding footprint of FinTech in Tanzania, driven by an increase in digital adoption and innovation in financial services (Statista report, 2024).

Additionally, the Bank of Tanzania (BOT) introduced the Tanzania Instant Payment System (TIPS) to facilitate real-time payments across different financial providers, addressing issues such as liquidity management and operational inefficiencies (BOT, 2020). With further advancements, such as the modernization of the Tanzania Interbank Settlement System (TISS) and the FinTech Regulatory Sandbox 2023, the groundwork has been laid for a deeper investigation of a bank's FinTech influence on banking efficiency based on their categories. As an emerging economy, Tanzania's adoption of FinTech presents opportunities to improve the banking sector (Makina, 2019). Many banks have initiated their FinTech projects or partnered with firms such as mobile network operators (MNOs) to integrate financial services (Koloseni & Mandari, 2024). These collaborations enable smooth transactions between Fintech platforms and commercial banks.

Existing studies of bank efficiency often focus on bank-specific and macroeconomic factors. For example, Temba *et al.* (2023) and Lotto (2019) examine governance and liquidity in Tanzanian banks, but their findings do not account for the growing role of FinTech. Additionally, studies such as Sajid *et al.* (2023) and Isik and Uygur (2021) emphasize the role of FinTech in enhancing operational efficiency through automation and resource optimization. Most of the studies conducted in Africa, such as Shah *et al.* (2022), Banya and Biekpe (2018), Lotto (2019 and 2018), and Lema (2017), analyzed banks' efficiency using data envelopment analysis (DEA) but did not focus on bank FinTech to evaluate its influence on banks' efficiency. Few studies have focused on the effects of competition between FinTech firms and bank performance.

This study addresses these gaps by constructing a bank FinTech index for Tanzanian commercial banks, a novel contribution reflecting the level of FinTech development in this emerging economy. Unlike broader indices, this study develops a FinTech index tailored to Tanzanian banks' operational and technological dynamics, providing a localized perspective. The scarcity of research on FinTech integration and its efficiency implications was identified through a systematic review of the literature that revealed limited studies exploring FinTech's impact on bank categories in Tanzania. While previous research has highlighted general efficiency determinants, the role of FinTech remains underexplored. This gap forms the basis for this study's contribution, which examines how FinTech influences efficiency, measured by total factor productivity (TFP), across large, medium, and small banks. Additionally, this study examines the influence of control variables on bank efficiency to provide deeper insights into the drivers of bank efficiency in Tanzania. This study seeks to answer the following questions: How does the bank FinTech index influence efficiency in Tanzanian commercial banks? Does this effect vary by bank category? What are the broader implications for the banking sector, particularly regarding the influence of the bank control variables on bank efficiency?

The results show that large banks benefit significantly from the FinTech index, enhancing efficiency through advanced technology adoption. In contrast, medium and small banks face challenges in scaling FinTech, negatively affecting their efficiency. Key factors such as the capital adequacy ratio and loan-to-deposit ratio influence efficiency differently across bank sizes. Nonperforming loans consistently reduce efficiency, emphasizing the need for robust risk management. By combining theoretical models with data-driven validation, the study offers actionable insights for policymakers and managers to optimize FinTech integration and develop sustainable policies to enhance efficiency and resource utilization in Tanzania's banking sector.

The remainder of the paper is structured as follows. Section 2 reviews the theoretical and empirical literature relevant to the study. Section 3 outlines the research methodology. Section 4 presents and discusses the results, and Section 5 concludes the study with key findings, recommendations, and suggestions for future research.

## 2. Literature Review

# 2.1. Theoretical Frameworks

This study examines the influence of the bank FinTech index on bank efficiency using intermediation theory, as proposed by Gorton and Winton (2003). Intermediation theory emphasizes banks' role as financial intermediaries, reducing transaction costs and addressing information asymmetries to facilitate fund flows between savers and borrowers, thereby fostering economic growth and development (Muda *et al.*, 2021; Das Gupta *et al.*, 2021). The emergence of FinTech offers transformative potential to traditional banking intermediation by introducing innovative technologies that streamline operations, improve credit risk assessment, and enhance customer

experiences. These technologies ranging from mobile banking and artificial intelligence-driven credit scoring to blockchain platforms, among others, significantly reduce operational costs, improve risk management, and expand financial accessibility (Wang *et al.*, 2021). While intermediation theory offers a strong foundation for understanding the economic role of banks, it does not fully capture the dynamic and disruptive nature of FinTech. Technological innovations often extend beyond traditional banking frameworks, creating new forms of financial intermediation. This study broadens the theoretical scope by integrating Innovation Diffusion Theory (IDT) and the Technology Acceptance Model (TAM) to address these limitations.

Innovation Diffusion Theory (IDT), as proposed by Rogers *et al.* (2014), explains the adoption of innovations based on factors such as perceived relative advantage, compatibility with existing systems, complexity, trialability, and observability. In the context of FinTech, IDT highlights that banks adopt technologies like mobile banking and artificial intelligence analytics when their benefits, such as enhanced efficiency, cost reduction, and improved customer satisfaction, are perceived to outweigh potential challenges (Adewumi *et al.*, 2024). By focusing on these attributes, IDT provides a framework for understanding the rate and drivers of FinTech adoption within banks.

The Technology Acceptance Model (TAM), as proposed by Davis (1989), complements IDT by addressing behavioral and organizational factors influencing technology adoption. TAM identifies perceived ease of use (PEOU) and perceived usefulness (PU) as key drivers of acceptance. In banking, these factors are critical in determining how FinTech solutions are integrated into operations and used to enhance efficiency and customer engagement (Latreche et al., 2024). For instance, technologies perceived as user-friendly and beneficial are more likely to be embraced by bank staff and customers, accelerating their impact on bank profitability and operational efficiency (Isiaku & Adalier, 2024). This study captures the complex relationship between FinTech development and bank efficiency by integrating intermediation theory, IDT, and TAM. Intermediation theory provides the economic rationale, while IDT and TAM address organizational and behavioral dimensions. This comprehensive approach aligns with recent studies by Cheng (2020) and Odoom and Kosiba and offers actionable insights for banks and policymakers. It highlights how technological advancements reshape traditional banking operations and foster greater efficiency, particularly in emerging markets like Tanzania.

## 2.2. Empirical Review

Empirical studies examining financial technology's (FinTech) impact on bank efficiency have yielded various findings across regions and methodologies. The FinTech index is a composite measure capturing the extent of financial technology adoption by banks, encompassing dimensions like mobile banking, blockchain technology, artificial intelligence-driven credit scoring, digital payments, and online lending platforms (Wang et al., (2021). The relationship between FinTech and bank efficiency has been extensively explored, with most studies indicating positive impacts on various dimensions of bank performance (Lee et al., 2023; Chen et al., 2021). FinTech-related initiatives, often incorporated as independent variables or efficiency scores in econometric modeling, demonstrate the potential of technology to enhance banking operations. Das Gupta et al. (2021) developed a weighted FinTech index and found a positive correlation with operational efficiency in South Asian banks.

The rapid evolution of financial technology has highlighted the need for comprehensive indices to measure FinTech adoption and its impact on the banking sector. Existing efforts, such as Cheng and Qu'sa's (2020) banking FinTech index, utilized web crawlers and word frequency analysis to assess FinTech adoption broadly but lacked focus on sector-specific operational measures. Similarly, Cao et al. (2024) employed the Baidu search engine to track financial technology-related keywords and the efficiency in the Chinese banking sector, providing robust data-driven insights but without directly linking these trends to bank efficiency measures like cost reduction or customer satisfaction. Deng et al. (2021) further developed a FinTech index derived from municipal digital financial indices, offering a broader perspective on urban FinTech development but failing to capture bank-specific performance impacts.

This study's constructed Bank FinTech Index addresses these gaps by focusing on sector-specific metrics such as operational integration, technological adoption, and product innovation. Unlike prior indices, it explicitly connects FinTech adoption to bank efficiency outcomes, including cost efficiency, resource allocation, and enhanced customer service. This targeted approach offers a practical tool for analyzing the operational impact of FinTech on banking institutions. Empirical evidence underscores the importance of such tailored indices, demonstrating that effective FinTech integration can reduce transaction costs, improve risk management, and optimize resource utilization, making this Bank FinTech Index a valuable addition to the field.

Wang et al. (2021) also explored FinTech's influence in China by combining DEA and GMM to assess improvements in total factor productivity (TFP) within commercial banks. Their findings highlight that the degree of technological application significantly influences productivity outcomes, pointing to the importance of strategic technology adoption in achieving efficiency gains. In contrast, Khan et al. (2024), using GMM to study 59 developing nations, revealed that FinTech integration initially hampers bank efficiency due to adaptation challenges but leads to efficiency improvements as adoption matures. This non-linear relationship underscores the complexities of Fintech adoption, particularly in emerging markets.

Bank efficiency is typically measured using methodologies like Data Envelopment Analysis (DEA), Stochastic Frontier Analysis (SFA), and Total Factor Productivity (TFP), linking FinTech adoption to cost reductions, improved risk management, and customer engagement (Prakash et al., 2021). Chen et al. (2021) reported increased cost efficiency in Chinese banks with higher FinTech index scores using DEA. For instance, Liao (2023) employed a DEA-based model to assess the impact of FinTech on bank efficiency, revealing no significant improvements over traditional branches' practices. However, initial high implementation costs and varying impacts across bank sizes remain challenges (Ogbonna et al., 2023).

Collaborative efforts between FinTech firms and banks have been shown to enhance efficiency (Klus et al., 2019). Ntwiga (2020) emphasizes that such partnerships allow banks to reach customers more effectively through digital channels, reducing acquisition costs compared to traditional methods. Similarly, studies by Thakor (2020) and Wirtz et al. (2023) demonstrate that FinTech offers digital self-service options and personalized experiences to lower customer servicing costs and improve efficiency. Advanced technologies, such as data analytics and artificial intelligence (AI), enable banks to understand customer needs better, fostering targeted cross-selling opportunities and improving efficiency (Raj et al., 2024; Chen et al., 2021).

In East Africa, Ky et al. (2019) use a fixed-effects model to examine mobile money as proxies for FinTech, revealing a positive relationship between these innovations and bank efficiency. Similarly, Ntwiga (2020), in a study of Kenyan banks using DEA and a fixed-effects model, demonstrated that Fintech collaboration significantly reduced intermediation costs and increased operational scale, albeit with diminishing returns to scale. These studies highlight the transformative role of mobile money and collaborative Fintech solutions in enhancing efficiency, particularly in resource-constrained settings.

Empirical studies in Tanzania have focused on governance and operational factors influencing bank efficiency. Using a random-effects regression model, Lotto (2019 and 2018) found that bank liquidity and capital adequacy positively influence operating efficiency. Through multiple linear regression, Temba et al. (2023) highlighted the negative influence of corporate governance variables, such as board gender diversity and governance control, on the efficient use of equity and liquidity. By employing DEA and regression analysis, Soud and Aypek (2020) find that board size and

composition significantly impact efficiency, emphasizing the role of governance in

These studies provide diverse insights into the interplay between Fintech, governance, and operational efficiency. However, gaps remain in understanding how FinTech influences efficiency across different bank categories, such as large, medium, and small banks, particularly in emerging economies such as Tanzania. This underscores the need for further research to examine the influence of FinTech integration within various bank contexts and across different bank sizes.

A literature review indicates a research gap in evaluating the relationship between a bank's efficiency and the bank FinTech index in an emerging market such as Tanzania. The conflicting results of previous studies have led this study to test two key hypotheses. First, (H1) suggests that the FinTech index positively influences Tanzanian banks' efficiency. Second, (H2) the influence of the bank FinTech index on efficiency varies across different bank categories, including large, medium, and small banks. These hypotheses offer a comprehensive view of the factors driving bank efficiency among Tanzanian banks.

### 3. Methodology

# 3.1. Data and Study Design

shaping operational outcomes.

This study investigated the influence of the bank FinTech index on the efficiency of commercial banks in Tanzania. Grounded in a positivist research philosophy, this study empirically assesses the relationship between the bank FinTech index and bank efficiency. This study draws on a balanced panel dataset covering 2010 to 2021, with data obtained from the Bank of Tanzania Supervision Information System (BSIS), aggregating annual audited financial reports from commercial banks. Macroeconomic data were sourced from the National Bureau of Statistics (NBS), the Bank of Tanzania (BOT), and World Bank indicators.

A purposive sample of 30 commercial banks was selected from an initial pool of 36 banks based on their continuous operation during the study period and data availability. Narrowing the study to commercial banks allows for a more focused, relevant, and comparable analysis of the relationship between Fintech development and bank efficiency. The year 2010 was chosen as the starting point because of the rapid adoption of FinTech, beginning in 2008, transformed banking channels, with 2010 marking a critical period of growth in mobile and internet banking, highlighting the early stages of FinTech integration in the banking sector (BOT, 2019). The Bank of Tanzania classifies commercial banks into three peer groups based on total assets: Peer group 1 includes the most prominent banks with assets between TZS 500 billion and TZS 99.999 trillion; peer group 2 comprises medium-sized banks with assets ranging from TZS 200 billion to TZS 500 billion; and peer group 3 consists of smaller banks with assets between TZS 30 billion and TZS 200 billion (BOT, 2018). Banks were categorized as large, medium, and small based on total assets to account for heterogeneity in resources, technological capabilities, and operational strategies influencing FinTech integration. Large banks often have a greater capacity for advanced FinTech solutions, while medium and small banks may face resource constraints but exhibit unique adoption dynamics. This categorization enables analysis of whether the Bank FinTech Index's impact varies by size, providing targeted insights for policy and management decisions (Lee et al., 2021). Separate regressions were performed for each group of ten (10) banks to account for potential size-related differences that could affect model estimates, following the Bank of Tanzania's asset-based classification guidelines.

This study applied a two-step System Generalized Method of Moments (GMM) estimator to assess the influence of the bank FinTech index on bank efficiency (Medhioub & Boujelbene, 2024; Zhao *et al.*, 2022). A panel research design was employed to explore the trends in bank efficiency over time about the bank FinTech index. This

design is well suited for analyzing temporal changes, accounting for individual bank heterogeneity, and enhancing the statistical robustness of the results (Yitayaw, 2021). The advanced econometric approach addresses potential endogeneity and unobserved bank-specific effects, offering robust insights into the dynamic relationship between a bank's FinTech index and efficiency in Tanzania's banking sector (Roodman, 2009).

# 3.2. Total Factor Productivity and the Efficiency of Banks

This study employs total factor productivity (TFP) as a proxy variable to measure the efficiency of banks. TFP was calculated using the Data Envelopment Analysis (DEA) Malmquist method (Zhu et al., 2021; Li et al., 2021; Ferreira, 2020). This non-parametric technique is ideal for analyzing multiple inputs and outputs, aligning with banking operations' complexity (Boot et al., 2021). Measuring bank efficiency presents challenges because banks do not produce physical goods. Two primary approaches are used: the production method (which focuses on services such as account management and transactions) and the intermediary method, which views banks as intermediaries that transfer funds between savers and borrowers (Wang et al., 2021a). Regardless of whether the asset or intermediary method is adopted, production remains the key factor in maximizing income and shareholders' wealth (Wang et al., 2021a).

This study focuses on banks' production outcomes. Using the production method, the DEA Malmquist analysis incorporates labor costs, deposits, and total capital as inputs (Gökgöz et al., 2024; Maradin et al., 2021; Cheriye, 2020). Labor costs include all forms of employee compensation; deposits represent total customer funds held by the bank; and total capital covers both equity and debt capital (Gökgöz et al., 2024). The outputs selected for the analysis include profit, loans, and interest income, which represent net income, lending activity, and the bank's capacity to earn interest, respectively (Abedifar et al., 2018; Alhassan & Tetteh, 2017). Previous studies have successfully used TFP to evaluate bank efficiency in various contexts. For example, (2024) employed TFP to analyze productivity in emerging markets, while Maradin et al.) demonstrated its applicability in assessing efficiency during periods of financial transformation. Similarly, Alhassan and Tetteh (2018) highlighted the utility of TFP in linking operational outcomes to profitability and shareholder value in the banking sector.

This study assesses total factor productivity in commercial banks by applying the DEA Malmquist method, thereby analyzing the influence of the bank FinTech index on bank efficiency (Chen, 2021). All variables in the dependent, independent, and control groups were operationalized, as shown in Table 1.

## 3.3. The construction of the Bank FinTech Index

This study builds on the methodologies of Kharrat et al. (2024) and Wang et al. (2021a) to construct a FinTech index, detailing the key computational steps involved in this study. The FinTech data from banks were gathered using an Excel checklist sent to the banks, which featured nine (9) distinct FinTech dimensions and thirty-three (33) associated FinTech applications, as detailed in Appendix 1. The nine (9) dimensions and corresponding thirty-three (33) FinTech applications were adopted from the study of Wang et al. (2021) and customized based on their relevance in the Tanzanian banks context. The Banks were asked to report the availability of each application per dimension annually, assigning a value of one (1) for available and zero (0) for unavailable FinTech applications each year.

Before calculating the FinTech index, diagnostic tests were conducted to ensure adequate sample size. The results in Table 2 confirm that the scale analysis revealed strong internal consistency and suitability for further statistical analyses. The average inter-item covariance of 0.1177 indicates a positive relationship among the 33 items, suggesting that they measure related aspects of the construct. A reliability coefficient

(Cronbach's alpha) of 0.9818 reflects satisfactory internal consistency, implying that the items are highly correlated and measure the same underlying concept. However, this high reliability may indicate some redundancy, suggesting that certain items overlap. The Kaiser-Meyer-Olkin (KMO) measure of 0.9472 indicates sufficient sampling adequacy, confirming that the dataset is well-suited for factor analysis. These findings suggest that the scale is robust for measuring the intended construct, and further analysis, such as factor analysis, helps identify the key underlying dimensions (Durak et al., 2024; Anita et al., 2023; Theiri & Alareeni, 2023; Van Phuc Le & Nguyen, 2022; Singh et al., 2020). With the inclusion of thirty-three (33) FinTech applications, these results confirm that the data are suitable for planned factor analysis. The results of the KMO and Cronbach's reliability tests are shown in Table 2.

Variable	Operationalisation	Measurement	Empirical Evidence	
Independent Variable	The author owns FinTech index computation.	Composite	Katsiampa et al., (2022),	
FinTech Index (FTI)		Index	Wang et al., (2021b)	
Dependent Variables	Measures productivity changes as a proxy for	DEA	Zhu et al., 2021; Li et	
Total Factor Productivity (TFP <i>i</i> , <i>t</i> )	efficiency.		al., 2021;Ferreira, 2020	
Control Variables			Pierri and Timmer	
Return on Asset (ROA)	Profit before tax divided by average total assets	Ratio	(2020), Phan et al., 2020;	
Loan to Deposit Ratio (LDR)	Total loans divided by total deposits	Ratio	Wang et al. (2021b)	
Capital Adequacy ratio (CAR)	Total equity divided by total assets	Ratio	Dwivedi et al., (2021),	
Non-Performing Loans (NPLs)	Total NPL divided by gross total loans	Ratio	Haabazoka (2019),	
Bank Size	Logarithm of total assets		Ozili, (2021a&b),	
Inflation (INF)	Annual inflation rate (measured as a percentage	Value	(Msomi, 2022),	
Gross Domestic Product	change in consumer price index) of a country	Ratio	Singh et al., 2021),	
(GDP)	(GDP) Annual GDP growth rate (%) of a country		(Sinaga et al., 2020),	
			(Kartikasary et al.,	
			2020), Ahmed et al.,	
			(2021)	

Table 1. Operationalisation of Research Variables

*Note*: Source: Author's own compilation and literature review (2024)

Table 2. KMO and Cronbach's Test Results

Test scale = mean (unstandardized items)				
Average interitem covariance:	0.1176			
Number of items in the scale:	33			
Scale reliability coefficient:	0.9818			
Kaiser-Meyer-Olkin measure of sampling adequacy (KMO)	0.9472			

Note. Source: Survey Data (2024)

To manage the extensive range of FinTech applications, principal component analysis (PCA) was used to simplify the dataset by identifying key patterns among the 33 FinTech applications (Kharrat et al., 2024; Zheng et al., 2024; Rizvi et al., 2024; Aboojafari & Dehghani, 2022). PCA is a multivariate technique that reduces data dimensionality by analyzing several interconnected quantitative variables. This process identifies the principal components that highlight important factors within the FinTech dimensions, providing a clearer and more insightful representation. Similar approaches were successfully applied in previous studies by Zheng et al. (2024) and Kharrat et al. (2024). PCA allows data compression and simplification while retaining crucial information (Rahman, 2024).

Several key steps were undertaken to apply PCA to compute the FinTech index. First, the covariance or correlation matrix was computed from the standardized dataset, which consisted of n observations and p variables, to capture the relationships between variables (Lasisi & Attoh-Okine, 2018). The covariance matrix was calculated as follows:

$$\sum = \frac{1}{n-1} X^T.$$
(1)

where X = standardized data with n observations, and  $X^{T}$  denotes the transpose of X.

Second, principal components were extracted from the covariance matrix using PCA. Each principal component is a linear combination of the original variables, capturing the maximum variance, with subsequent components explaining progressively less variance (Peres & Fogliatto, 2018). The principal components are computed as follows:

$$PC_k = XV_k.$$
 (2)

where:  $PC_k$ = principal components, V = matrix of eigenvectors obtained from the covariance matrix, and  $V_k$ = Matrix of the first k eigenvectors corresponding to the largest eigenvalues.

The third step involves computing scores for each principal component for every bank, representing the data projection onto these components (Dong & Qin, 2018). If  $X_i$  represents the data for bank *i* and  $PC_k$  is the *k*-th principal component, the score for bank *i* on component *k* is calculated as;

$$Score_{ik} = X_i * PC_k.$$
 (3)

Fourth, each principal component was evaluated to determine the variance. The first component explains the most variance, with subsequent components explaining progressively less variance. Components with eigenvalues greater than one were retained (Schreiber, 2021). Finally, combining the retained components, the composite bank FinTech index is computed (Zhao et al., 2023). This involves multiplying each component by its corresponding weight and then summing these weighted values. We compute the FinTech index as follows:

$$FI = \sum_{k=1}^{m} Score_{ik} \times W_k \,. \tag{4}$$

where  $W_k$  = Weight of each retained component  $PC_k$  and m = number of retained principal components

The resulting bank FinTech index offers a composite measure of FinTech development among Tanzanian commercial banks. This method is supported by literature from (Kharrat et al. (2024), Zheng et al. (2024), and Yao and Song (2023), highlighting PCA's effectiveness of PCA in reducing dimensionality and capturing essential variability in datasets. The constructed Bank FinTech Index aligns with established indices like the Peking University Digital Financial Inclusion Index (PKU-DFII) by leveraging digital data sources and methodologies such as keyword search and frequency analysis (Muganyi et al., 2022). However, it differs in its narrower focus on the banking sector, emphasizing operational integration, technological adoption, and product innovation. At the same time, the PKU-DFII takes a broader approach to digital financial inclusion across various financial services and geographies. This sector-specific focus makes the Bank FinTech Index more relevant for analyzing bank-level FinTech adoption and efficiency, though its limited scope may not capture broader financial inclusion trends (Khan et al., 2023).

# 3.4. Model Specification

This study uses panel data to explore the relationship between the bank FinTech index and the efficiency of commercial banks. Methods such as pooled ordinary least squares (OLS), fixed effects, and random effects models may not be suitable because of challenges such as endogeneity and unobserved individual effects (Sheraz et al., 2022; Roodman, 2009; Blundell & Bond, 1998; Arellano & Bover, 1995). To overcome these challenges, this study employs a two-step system Generalized Method of Moments (GMM), a robust estimation technique that addresses issues such as autocorrelation and heteroscedasticity (Blundell & Bond, 1998; Arellano & Bover, 1995). GMM can be categorized into different GMM and system GMM, each with one- and two-step variations. System GMM is generally more effective than difference GMM, and the two-step variant is particularly well suited for handling problems such as heteroscedasticity and autocorrelation, as applied by Khan et al. (2024) and Ogbonna et al. (2023).

Accordingly, this study adopts a two-step system GMM for its estimations, a method that has been successfully applied in similar research (Dasilas & Karanović, 2023; Sheraz et al., 2022; Ullah et al., 2021; Phan et al., 2020; Arminen & Menegaki, 2019). The linear representation of the dynamic GMM system model is as follows;

 $TFP_{i,t} = \alpha + \beta_1 FTI_{i,t} + \beta_2 CAR_{i,t} + \beta_3 SIZE_{i,t} + \beta_4 LDR_{i,t} + \beta_5 NPL_{i,t} + \beta_6 ROA_{i,t} + \beta_7 GDP_t + \beta_8 INF_t + \epsilon_{i,t}.$ (5)

To ensure our results are valid, the interaction terms test whether the relationship between the FinTech Index and efficiency is consistent across different bank sizes (large, medium, and small) was included in the estimation model for the robustness test. This checks the model's ability to capture differential impacts, which adds depth and validity to the findings. The model indicating the interaction term of the FinTech Index is as follows:

$$\begin{aligned} \text{TFP}_{i,t} &= \alpha + \beta_1 \text{FTI}_{i,t} + \beta_2 (\text{FTI}_{i,t} \times \text{Large}_{i,t}) + \beta_3 (\text{FTI}_{i,t} \times \text{Medium}_{i,t}) + \beta_4 (\text{FTI}_{i,t} \times \text{Small}_{i,t}) + \beta_5 \text{CAR}_{i,t} + \beta_6 \text{SIZE}_{i,t} + \beta_7 \text{LDR}_{i,t} + \beta_8 \text{NPL}_{i,t} + \beta_9 \text{ROA}_{i,t} + \beta_{10} \text{GDP}_t + \beta_{11} \text{INF}_t + \epsilon_{i,t} , \end{aligned}$$

$$\end{aligned}$$

where: All variable definitions for equations one and two are detailed in Table 1 and  $\varepsilon i,t$ = Error term, capturing the combined effect of all other factors not explicitly included in the model that might influence a bank's TFP. Large, medium, and small banks are the categorization of banks, as described in section 3.1 of this paper.

# 3.5. Descriptive statistics

Table 3 presents the descriptive results of the variables used to examine the influence of the bank FinTech index on the efficiency of banks across different categories of commercial banks in Tanzania (All Banks, Large, Medium, and Small Banks). The FinTech Index (FTI) has an overall mean of 1.918 with a standard deviation of 2.175. Large banks exhibit a higher mean FTI of 2.544 (Std. Dev. = 2.162), which reflects a more advanced deployment of FinTech compared with medium banks (Mean = 1.476, Std. Dev. = 2.04) and small banks (mean = 1.736, Std. Dev. = 2.194). This finding suggests that larger banks are better positioned to invest in and leverage financial technologies. The detailed descriptive results on the variables used are presented in Table 3.

	All Ba	anks (360)	Large I	Banks (120)	Mediun	n Banks (120)	Small I	Banks (120)
Variable	Mean	Std. Dev.	Mean	Std. Dev.	Mean	Std. Dev.	Mean	Std. Dev.
TFP	1.09	0.531	1.042	0.346	1.096	0.497	1.132	0.692
FRI	1.918	2.175	2.544	2.162	1.476	2.04	1.736	2.194
CAR	16.575	11.18	13.169	2.868	14.375	7.983	22.18	16.032
LDR	81.565	48.77	67.003	16.475	82.143	23.422	95.55	77.1
ROA	0.129	4.81	2.201	1.957	-0.365	5.139	-1.449	5.699
GDP	5.783	1.49	5.783	1.494	5.783	1.494	5.783	1.494
INF	6.583	3.778	6.583	3.789	6.583	3.789	6.583	3.789
In Size	11.465	0.841	12.113	0.346	11.445	0.342	10.837	1.038
NPL	9.215	12.48	6.614	5.16	11.312	15.97	9.719	13.286

Table 3. Descriptive Statistics Results

Note. Source: Survey Data (2024)

For total factor productivity (TFP), the average for all banks is 1.09, with small banks showing the highest mean TFP (1.132) and large banks the lowest (1.042). The FinTech Index (FTI) shows the highest average for large banks (2.544), while medium banks have the lowest (1.476). Capital Adequacy Ratio (CAR) is notably higher in small banks (22.18), reflecting stronger capital buffers, while large banks have the lowest CAR (13.169). The Loan-to-Deposit Ratio (LDR) is highest for small banks (95.55), indicating higher lending activities relative to deposits, whereas large banks exhibit a lower LDR (67.003). Return on Assets (ROA) reveals profitability disparities, with large banks having a positive average ROA (2.201), while medium and small banks show negative ROA values, reflecting lower profitability or potential losses. Macroeconomic variables such as GDP and inflation remain consistent across all categories. Large banks have the highest average logarithmic size (In Size 12.113), and small banks have the smallest (10.837). Non-Performing Loans ratios are highest in medium banks (11.312), suggesting higher credit risk, while large banks have a relatively lower NPL ratio (6.614). The high standard deviations, especially for variables such as FTI, CAR, LDR, ROA, and NPL, indicate significant variability across banks, reflecting differences in size, risk exposure, and operational strategies. Small banks show the highest variability, suggesting more performance and risk management volatility than large and medium banks.

#### 3.6. Diagnostic Tests

Before estimating the system GMM, several diagnostic tests were conducted to ensure the robustness of the analysis. First, the data were examined for potential multicollinearity among independent variables to avoid distortions in the results. Additionally, tests were performed to address the unique characteristics of the panel data, including cross-sectional dependence and stationarity. Hansen and Sargan tests were also employed to validate the instruments used in the model, ensuring that they were not correlated with the error term, thereby providing unbiased and reliable estimations.

### 3.6.1. Pairwise Correlations and Multicollinearity Results

Table 4 presents the pairwise correlations and Variance Inflation Factor (VIF) results for the bank FinTech index, control variables, and efficiency. The correlations ranged from weak to moderate, with no value exceeding 0.594, indicating a satisfactory level of variable distinction and minimal risk of multicollinearity. This suggests that multicollinearity is not a significant issue in our model, as the independent variables do not excessively inflate each other's variance owing to correlations (Asongu et al., 2021). This approach was consistent with the methodologies employed by Kashif et al. (2024) and Aggarwal et al. (2023). Multicollinearity was further assessed using VIF values, and Table 5 shows that all variables had VIF values below 3 and corresponding tolerance values (1/VIF) above 0.433. These results confirm that multicollinearity is not a significant concern in our model, as no independent variable's variance is substantially inflated by its correlation with other variables (Asongu et al., 2021). This approach aligns with the findings of Kashif et al. (2024) and Aggarwal et al. (2023), who reported no multicollinearity using the VIF in their analyses. The detailed results of pairwise correlations and variance inflation factors are presented in Table 4.

3.6.2. Cross-Sectional Dependency and Panel Unit Root Tests

Table 5 presents the results of the cross-sectional dependency and panel unit root tests, using the Breusch-Pagan LM, Pesaran CD, and Friedman tests, as outlined by (Li et al., 2024; Antwi & Kong, 2023; Shahbaz et al., 2023). Detailed cross-sectional dependency and panel unit root test results are presented in Table 5.

Prior to conducting the two-step system GMM analysis, this study carefully assessed the data for cross-sectional dependence, a common feature in panel data, where a single shock, such as regulatory changes, can affect multiple banks (Khalid & Shafiullah, 2021). Table 4 presents the results of cross-sectional dependence tests,

including the Breusch-Pagan LM, Pesaran CD, and Friedman tests, following the methodologies of Rana et al. (2024) and Shahbaz et al. (2023). As the significance test statistics show, the results indicate significant cross-sectional dependence across all variables. Consequently, this study employs second-generation panel unit root tests, which are necessary for evaluating the stationarity of variables in the presence of cross-sectional dependence, as first-generation tests may otherwise produce misleading results (Eibinger et al., 2024). Table 4 also presents the results of the Cross-sectionally Augmented IPS (CIPS) panel unit root test, which integrates the augmented Dickey (ADF) and Im-Pesaran-Shin tests. The CIPS test findings show that CAR and LDR achieve stationarity after first differencing, indicating that they are integrated in order one. These results are robust to cross-sectional dependence and provide more consistent and reliable outcomes (Eibinger et al., 2024; Rana et al., 2024; Shahbaz et al., 2023).

Table 4.	Correlations	and Multico	ollinearity	Results

Variables	TFP	FTI	CAR	LDR	ROA	GDP	INF	In Size	NPL
TFP	1								
FTI	0.007	1							
CAR	0.033	-0.061	1						
LDR	0.059	0.05	0.594***	1					
ROA	-0.097*	0.077	-0.447***	-0.267***	1				
GDP	-0.018	-0.554***	0.03	-0.013	0.038	1			
INF	-0.071	-0.537***	0.046	-0.072	-0.07	0.221***	1		
lnSize	-0.027	0.314***	-0.648***	-0.430***	0.493***	-0.106**	-0.165**	1	
NPL	-0.036	0.148**	0.024	0.082	-0.237***	-0.131**	-0.217***	-0.014	1
VIF		2.21	2.31	1.61	1.49	1.49	1.48	2.24	1.15
1/VIF		0.452	0.433	0.623	0.669	0.669	0.675	0.447	0.872
Mean VIF		1.75							

Note. Standard errors: \*\*\* p<0.001, \*\* p<0.05, \* p<0.1

# Table 5. Cross-Section Dependency and Panel Unit Root Test Results

Cross-section dependency (CD)			Cross-sectionally augmented IPS (CIPS) panel unit root			
	Breusch-Pagan LM					
	test	Pesaran CD test	CIPS J	CIPS panel unit root		est difference
						Constant and
Variables	Test statistics	Test statistics	Constant	Constant and trend	Constant	trend
TFP	585.518***	1.774*	-4.400***	-4.349***		
FTI	3503.384***	58.334***	-2.792***	-3.173***		
CAR	1331.610***	4.158***	-1.919	-2.311	-3.039***	-3.207***
lnSize	3344.896***	54.755***	-2.498***	-3.222***		
ROA	770.613***	4.269***	-2.011	-2.770**		
GDP	5220***	72.250***	2.610***	1.7		
INF	5220***	72.250***	2.610***	1.7		
NPL	1252.372***	10.324***	-2.701***	-2.773**		
LDR	1175.317***	17.375***	-1.974	-1.837	-2.770***	-2.965 ***

*Note.* Standard errors in parentheses: \*\*\* p<0.001, \*\* p<0.05, \* p<0.1

3.6.3. Hansen and Sargan Test

Table 7 shows the diagnostic test results for the Generalized Method of Moments (GMM) model, which confirms its validity through several key indicators. First, the first-order autocorrelation AR (1) test shows significant p-values of 0.015 and 0.000, indicating the expected presence of first-order serial correlations in the differenced residuals. Importantly, the second-order autocorrelation AR (2) test results, with p-values of 0.092,

0.252, 0.426, and 0.576, show no significant second-order autocorrelation, which is a critical condition for model validity (Bunje et al., 2022; Hsieh and Lee, 2020). The Hansen test for over-identifying restrictions, with p-values of 0.403, 0.113, 1.000, and 1.000, confirms the validity of the instruments, as the null hypothesis of no correlation between the instruments and the error term cannot be rejected (An et al., 2023). Similarly, the Sargan test showed p-values of 0.109, 0.199, 0.358, and 0.043, mostly indicating instrument validity (Ma and Lv, 2023). Additionally, the difference-in-Hansen Test validates the extended model specification. Despite its robustness, system GMM faces limitations like instrument proliferation, mitigated by collapsing instruments and restricting lag ranges, and assumptions of exogeneity and initial conditions, addressed through careful instrument selection and robustness checks (Roodman, 2009; Blundell and Bond, 1998). Overall, these tests suggest that the GMM model is well specified, and the instruments used are valid, making the results reliable for further interpretation.

## 4. Results and Discussion

## 4.1. Influence of a Bank FinTech Index on the Efficiency of Commercial Banks

Table 7, panels 1 to 4, presents the regression results on the influence of the bank FinTech index on the efficiency of commercial banks across all banks, as well as for large, medium, and small banks in Tanzania. Using panel data and two-step GMM system estimators, the analysis shows a strong and statistically significant positive relationship between the bank FinTech index and efficiency for all banks and large banks. However, the results indicate a negative influence of the FinTech index on the efficiency of medium and small banks.

The results provide crucial insights into the relationship between the bank FinTech Index (FTI) and bank efficiency in Tanzania. For all banks, the bank FinTech index positively and significantly influences efficiency, with a coefficient of 0.100 (p<0.001). This implies that an increase in FinTech development, as measured by the bank FinTech index, leads to enhanced operational productivity across the banking sector. The positive relationship between the bank FinTech index and TFP suggests that banks investing in FinTech experience streamlined operations, reduced transaction costs, and improved service delivery. These results highlight the importance of FinTech in driving overall bank efficiency, especially as digital platforms and automation have become more integral to banking operations. The study conducted by Lee et al. (2023) on 74 Chinese commercial banks from 2012 to 2019 revealed that different types of commercial banks were impacted by FinTech to varying degrees.

For Tanzanian banks, the possible increase in the deployment of mobile banking, digital payment platforms, and automated banking services is revolutionizing service delivery and customer experience (Koloseni, 2021). This transformation is especially relevant in countries where access to traditional banking services is limited, particularly in rural areas (George *et al.*, 2024). The introduction of digital financial platforms has extended banks' reach, allowing them to cater to previously underserved populations. As FinTech integration continues to increase, Tanzanian banks can leverage these innovations to improve operational efficiency, increase customer satisfaction, and maintain a competitive edge in an increasingly digital economy.

For large banks, the positive effect of the bank FinTech index on TFP is even more pronounced, with a coefficient of 0.889 (p<0.001). Large banks, possibly benefiting from more resources, infrastructure, and human capital, are well positioned to leverage FinTech to improve their productivity significantly. This finding indicates that FinTech plays a vital role in enabling large banks to enhance their operational processes and maintain a competitive edge. By deploying advanced technologies, such as machine learning, mobile banking, artificial intelligence (AI), and big data analytics, large banks can efficiently manage data, improve customer services, and scale their operations, leading to continued productivity growth.

The findings also highlight the importance of FinTech for large banks in Tanzania, which are often better positioned to integrate advanced technologies and realize substantial productivity gains. Through their greater resources and infrastructure, large banks in Tanzania can implement FinTech solutions more effectively, benefiting from improved data analytics, automation, and risk management capabilities. This allows them to drive innovation, maintain high levels of operational efficiency, and offer customers a wider range of financial products. The same results were reported by Lee *et al.* (2021) and Zhao *et al.* (2022), who concluded that FinTech development and FinTech innovations improve the cost efficiency of large banks in China and enhance the technology banks use.

**Table 6.** System GMM Results for all Banks and by Bank Categories with the inclusion of Interaction Effect

VARIABLES	(1)	(2)	(3)	(4)	(5)
	All Banks	Large Banks	Medium Banks	Small Banks	Interaction
		-			Effect (FTI)
L.TFP	-0.274**	1.815**	4.969*	-0.594**	-0.234***
	(0.128)	(0.790)	(2.718)	(0.274)	(0.071)
FTI	0.100***	0.889***	-3.811*	-0.899**	
	(0.022)	(0.240)	(2.287)	(0.416)	
FTI*Small					-0.138***
					(0.028)
FTI*Medium					-0.268***
					(0.061)
FTI*Large					0.088**
					(0.047)
CAR	-0.011**	0.052**	0.666**	-0.182***	-0.013***
	(0.005)	(0.021)	(0.333)	(0.049)	(0.004)
LDR	0.004*	0.012***	-0.174**	0.020***	0.005*
	(0.002)	(0.005)	(0.081)	(0.006)	(0.003)
ROA	-0.025*	0.270**	-0.606*	-0.090**	-0.018*
	(0.013)	(0.130)	(0.318)	(0.038)	(0.010)
GDP	0.053***	-0.343***	-0.818	-1.433**	0.036***
	(0.018)	(0.100)	(0.525)	(0.632)	(0.012)
INF	0.004	-0.065	0.005	-0.090	0.003
	(0.007)	(0.053)	(0.113)	(0.125)	(0.011)
lnBSize	-0.091	0.146*	43.642**	0.692	-0.223
	(0.153)	(0.080)	(22.002)	(2.118)	(0.289)
NPL	-0.015***	0.012	0.005	-0.014**	0.008
	(0.005)	(0.046)	(0.029)	(0.007)	(0.006)
Constant	1.883	0.000	-489.520**	6.693	3.335
	(1.840)	(0.000)	(246.344)	(21.877)	(3.588)
Observations	330	110	110	110	330
Number of banks	30	10	10	10	30
AR (1)	0.015	0.000	0.218	0.128	0.000
AR (2)	0.092	0.252	0.426	0.576	0.080
Hansen	0.403	0.113	1.000	1.000	0.258
Sargen	0.109	0.199	0.358	0.043	0.073
Number of Instruments	20.000	10	10	10	25.000

Note. Standard errors in parentheses: \*\*\* p<0.001, \*\* p<0.05, \* p<0.1

Conversely, medium banks show a negative and significant relationship between the bank FinTech index and TFP, with a coefficient of -3.811 (p<0.05), suggesting that increased FinTech deployment leads to decreased efficiency. This unexpected result could be due to challenges in implementing FinTech solutions, limited scalability, insufficient resources to manage such technologies, or substantial initial investment costs that cannot materialize in a shorter period. Medium banks may struggle with high implementation costs and the lack of expertise necessary to fully integrate and benefit from Fintech innovations (Stulz, 2019). The results highlight the need for medium banks to develop robust strategies, such as partnering with technology providers or outsourcing, to manage Fintech solutions and improve operational efficiency effectively. These results are consistent with the findings of Khan *et al.* (2024), who reported that FinTech initially decreases the efficiency of Chinese banks in the short run.

Similarly, FTI negatively affects TFP for small banks, with a coefficient of -0.899 (p<0.05). This suggests that small banks face even greater challenges in deploying and integrating FinTech technologies, likely because of resource constraints, limited technical expertise, and scalability issues. Small banks may struggle to generate adequate returns from their Fintech investments, as the costs associated with implementation and management may outweigh the benefits. A possible solution to overcome these challenges is to develop tailored, cost-effective, and scalable FinTech solutions that address their specific needs and limitations, potentially through partnerships or shared FinTech services. These results are consistent with the findings of Lee *et al.* (2023), who reported that FinTech impacts different types of commercial banks to varying degrees; urban and rural commercial banks are the most influenced, while joint-stock banks experience the least impact. FinTech initially decreases the efficiency of banks in the short run for Chinese banks.

However, this study's implications also highlight potential challenges for medium and small banks in Tanzania, where the benefits of FinTech integration may not be realized immediately. These banks may face resource limitations or technical barriers in successfully integrating FinTech solutions, potentially resulting in efficiency losses. Therefore, policies that support the capacity building of smaller banks, such as government-backed FinTech partnerships or technological subsidies, could help level the playing field and ensure that all banks benefit from FinTech-driven efficiency gains, regardless of size.

## 4.2. Influence of Control Variables on Efficiency of Banks

After analyzing the influence of the bank FinTech index on the efficiency of all banks and based on the categories, the study delved into analyzing the influence of bank-specific and macroeconomic factors to gain more insights into bank efficiency. The results in Table 7 are disaggregated across all banks and categorized into large, medium, and small banks, revealing distinct trends and implications for each category.

The results for capital adequacy ratio (CAR) across all banks, with a negative and significant coefficient of -0.011 (p<0.05), suggest that maintaining higher levels of capital adequacy may be associated with reduced efficiency. This negative relationship is possible because of the conservative lending practices that typically accompany higher capital reserves. When banks hold higher capital levels to meet regulatory requirements, they may adopt more risk-averse strategies, such as restricting lending to riskier borrowers or investing in growth opportunities. While these strategies ensure financial stability and regulatory compliance, they can also limit banks' ability to generate higher returns through profitable lending and investment activities, thereby reducing overall productivity. The same result was reported by Abba et al. (2018), who indicated that capital adequacy negatively influenced the efficiency of banks in Nigeria. In this context, a conservative approach to risk management and capital retention may constrain a bank's ability to innovate, expand, or exploit more lucrative market opportunities, which is reflected in the reduction of efficiency (Torre, 2020).

Conversely, for large banks, the results show a positive and significant relationship between the CAR and efficiency, with a coefficient of 0.052 (p<0.05). This indicates that large banks can leverage higher capital buffers to support, rather than hinder, operational productivity. Large banks have the advantages of broader resource bases, more diversified revenue streams, and greater access to capital markets. These factors allow firms to maintain high capital adequacy ratios without significantly compromising their lending capabilities or investment strategies. Rather than restricting growth, higher capital reserves enable large banks to engage in more complex transactions while maintaining financial resilience against economic shocks. This resilience allows them to invest in advanced technologies, improve customer service, and expand to new markets, all contributing to improved efficiency. These results are consistent with Siddique et al. (2022) and Dao (2020), who reported that capital adequacy positively influences the efficiency of large banks in South Africa and Vietnam. Moreover, large banks may use their capital reserves to explore innovative financing mechanisms such as digital banking and FinTech partnerships, further enhancing operational efficiency by reducing transaction costs and streamlining operations (Sajid et al., 2023; Zuo et al., 2023).

The findings for medium banks indicate that capital adequacy plays a pivotal role in driving long-term productivity, with a coefficient of 0.666 (p<0.05). This strong positive relationship suggests that for medium-sized banks, maintaining a higher capital adequacy ratio is beneficial and essential for their sustainability and growth. One possible explanation is that medium banks, which may not have the extensive resources or diverse revenue streams that larger banks enjoy, rely more heavily on strong capital buffers to manage risk effectively. Maintaining adequate capital reserves ensures that medium banks can absorb potential losses from loan defaults or market fluctuations, stabilizing their operations and enhancing productivity. Higher capital adequacy may also give banks financial flexibility to engage in strategic investments and expansion without overextending their resources.

Further, the results for small banks show a negative and significant relationship between capital adequacy and efficiency, with a coefficient of -0.182 (p<0.01). This suggests that, for small banks, maintaining higher levels of capital adequacy may cost their operational efficiency. Small banks typically have limited resources and face greater constraints in revenue generation. Thus, the requirement to hold higher capital reserves could limit their ability to lend or invest in growth opportunities. This capital constraint may reduce banks' capacity to engage in profitable lending activities, thereby increasing their productivity. Furthermore, small banks often operate in niche markets and may find it more challenging to compete with larger banks. Consequently, allocating a substantial portion of their resources to meet regulatory capital requirements might divert funds away from investments in technology, innovation, and customer service improvement, which are critical for enhancing efficiency in an increasingly competitive banking environment. These findings highlight the importance of tailoring capital adequacy requirements to banks' size and operational structure to ensure that regulatory frameworks support stability and efficiency across different bank categories.

The results for the Loan-to-Deposit Ratio (LDR) for the entire banking sector indicate that a higher LDR slightly improves efficiency (0.004, p<0.1), indicating that banks efficiently utilize their deposits to provide loans experience enhanced productivity. This effect is even more pronounced for large banks (0.012, p<0.001), which are better equipped to manage their larger loan portfolios. However, for medium banks, the relationship is negative (-0.174, p<0.05), suggesting that inefficiencies arise when banks overextend their lending capabilities relative to deposits. For small banks, the positive and significant effect (0.020, p<0.001) indicates that higher loan activity relative to deposits can enhance efficiency, although this may increase risk exposure.

The return on assets (ROA) for all banks indicates a negative and marginally significant relationship (-0.025, p<0.1), implying that higher profitability does not

necessarily translate to increased efficiency. This may be due to the possibility that banks focusing too much on returns may overlook operational improvements. However, for large banks, profitability and efficiency are positively linked (0.270, p<0.05), as these banks are likely to use their profits to fuel productivity-enhancing initiatives. Conversely, medium banks show a significant adverse effect (-0.606, p<0.1), suggesting that prioritizing profitability may reduce operational efficiency. Similarly, small banks show a negative and significant relationship (-0.090, p<0.05), indicating that balancing profitability with operational efficiency is challenging, likely because of resource constraints.

Non-Performing Loans (NPL), for all banks, indicates a significant negative relationship (-0.015, p<0.001) between NPLs and efficiency, meaning that higher levels of non-performing loans reduce bank productivity. This is consistent across the banking sector, as NPLs increase financial strain and operational inefficiency. Large and medium banks, however, do not exhibit significant impacts, suggesting they may have better mechanisms to manage NPLs. For small banks, the negative and significant coefficient (-0.014, p<0.05) shows that inefficiencies arise from higher NPLs, indicating weaker credit risk-management frameworks in smaller banks.

The results for Gross Domestic Product (GDP) growth for all banks have a positive and significant impact on efficiency (0.053, p<0.001), suggesting that a favorable macroeconomic environment drives demand for banking services, thereby improving productivity. However, for large banks, the impact is negative (-0.343, p<0.001), possibly because of the complexities in managing larger operations during periods of rapid economic expansion. Although the relationship is insignificant for medium-sized banks, the negative coefficient (-0.818) suggests they may struggle to fully capitalize on economic growth. Small banks face a significant negative effect (-1.433, p<0.05), highlighting their challenges in adjusting to macroeconomic changes, potentially because of their limited flexibility and risk management capabilities.

The results for Bank Size indicate that large banks show a marginally positive and significant effect (0.146, p<0.1), indicating that scale efficiency contributes positively to their operational performance. For medium banks, the relationship is highly significant and positive (43.642, p<0.05), implying that, as these banks grow, they can better utilize their resources to improve efficiency. However, small banks show no significant relationship, indicating that increasing size does not automatically translate into productivity gains.

#### 4.3. Robustness Check

To ensure the robustness and validity of the results, this study incorporates interaction terms between the Bank FinTech Index (FTI) and bank size categories (large, medium, and small) to account for potential differential impacts of FinTech adoption across these groups (Cheng & Qu, 2020). This approach enables a detailed examination of whether the relationship between FinTech and efficiency varies based on bank size, providing additional confidence in the findings. The results consistently demonstrate a significant positive relationship between the FinTech Index and efficiency for large banks, as evidenced by the positive and statistically significant interaction term (FTI Large = 0.088). This finding highlights the capacity of large banks to leverage advanced FinTech solutions effectively, owing to their superior resources and operational infrastructure. Conversely, the interaction terms for small and medium banks (FTI Small = -0.138, FTI Medium = -0.268) are negative and statistically significant, indicating challenges in translating FinTech adoption into efficiency gains. These challenges are likely driven by resource constraints, scalability limitations, and operational inefficiencies commonly observed in smaller banks (Wang et al., 2021a; Stulz, 2019). By integrating interaction terms into the regression model, the study ensures that the results are not overly generalized and reflect the complex dynamics of FinTech integration across different bank categories. This methodological enhancement reinforces the reliability and

credibility of the analysis, offering valuable insights for policymakers and practitioners seeking to tailor strategies for FinTech integration in the banking sector.

## 5. Conclusion, Recommendations, and Policy Implications

#### 5.1. Conclusion

In conclusion, this study examines the relationship between the bank FinTech index and efficiency in 30 Tanzanian commercial banks from 2010 to 2021, providing insights into the influence of FinTech development on bank performance. The findings reveal that the bank FinTech index has a significantly positive effect on efficiency across all banks, highlighting the role of FinTech in enhancing operational processes and service delivery. This influence is particularly evident in large banks, which are better equipped to integrate advanced technological solutions, improving their efficiency. By contrast, medium and small banks face challenges in effectively adopting and managing FinTech innovations, leading to a negative relationship between the FinTech index and their efficiency. These findings underscore the importance of scalability and resource availability in leveraging FinTech to improve performance. Furthermore, control variables such as capital adequacy and loan-to-deposit ratios significantly influence efficiency. Capital adequacy positively impacts large and medium banks. However, it hurts small banks, while the loan-to-deposit ratio improves efficiency in large and small banks but hinders efficiency in medium banks. Non-performing loans consistently correlate negatively with efficiency across all bank categories, emphasizing the importance of effective risk management.

#### 5.2. Recommendations and Policy Implications

The findings provide actionable recommendations for advancing FinTech adoption and enhancing bank efficiency in Tanzania by leveraging regulatory, technological, and collaborative advancements. Banks should recruit directors with expertise in FinTech and provide regular training on emerging technologies to support informed decision-making. Establishing FinTech innovation units and incentivizing employee-driven ideas can embed a culture of innovation. At the same time, collaboration with FinTech firms and leveraging the FinTech Regulatory Sandbox 2023 introduced by the Bank of Tanzania enables risk-free testing and optimized technology integration. Policymakers play a crucial role in initiatives like the Tanzania Instant Payment System (TIPS), promoting interoperability among commercial banks to enhance seamless digital transactions and financial inclusivity. Strengthening data protection frameworks under the Data Protection Act is vital to ensuring customer privacy and trust in digital services. Furthermore, partnerships between commercial banks and mobile network operators, such as those enabling M-Pesa and Tigo Pesa, can bridge mobile money platforms and traditional banking, expanding financial access and fostering mobile banking innovations. These strategies establish a robust and inclusive foundation for FinTech-driven transformation in Tanzania's banking sector.

#### 6. Study Implications, Limitations, and Areas for Further Studies

#### 6.1. Theoretical implication

This study contributes to advancing theoretical frameworks by integrating Intermediation Theory, Innovation Diffusion Theory (IDT), and the Technology Acceptance Model (TAM) to examine how financial technology (FinTech) enhances bank efficiency in Tanzania. It deepens Intermediation Theory by showing how FinTech reduces transaction costs, improves resource allocation, and expands financial inclusion, particularly benefiting large banks equipped to integrate advanced technologies. It also extends the theory by highlighting challenges for medium and small banks in adopting

FinTech, emphasizing the importance of resources and strategic implementation. Incorporating IDT, the study illustrates how banks adopt FinTech based on perceived benefits like cost reduction and compatibility with existing systems, enriching the theory by contextualizing FinTech adoption in emerging markets. The TAM framework complements this by addressing behavioral and organizational dimensions, emphasizing that technologies perceived as easy to use and valuable are more readily embraced, enhancing efficiency and profitability. By integrating these three frameworks, the study provides a comprehensive view of how FinTech transforms banking operations, aligning with recent research by *Sunardi et al.* (2022) and Fianto *et al.* (2021) and offering actionable insights for banks and policymakers in emerging economies.

## 6.2. Limitations and Area for Further Studies

This study had some limitations that warrant consideration. First, the analysis focuses primarily on the influence of the Bank FinTech Index on efficiency. However, it does not extensively explore the impact of other technological advancements or external disruptions, such as changes in regulatory frameworks or global shocks. Additionally, the study mainly addresses banks in Tanzania, limiting the generalizability of the findings to other regions with different financial infrastructures. Future studies should evaluate the long-term impact of Fintech on bank efficiency by including regulatory changes and global shocks across different countries.

**Acknowledgment:** We would like to sincerely thank the Bank of Tanzania, Bank of Tanzania Academy, and Moshi Cooperative University, particularly the Department of Microfinance and Banking, for their invaluable support and assistance throughout this study. Their contributions, including data provision and expert guidance, have successfully completed this research.

**Author contributions:** Conceptualization, O.J.A. and Y.K.; methodology, O.J.A.; software, O.J.A.; validation, O.J.A., Y.K., and L.M.; formal analysis, O.J.A.; investigation, O.J.A.; resources, O.J.A.; data curation, O.J.A.; writing original draft, O.J.A.; review and editing, Y.K. and L.M.; visualization, O.J.A.; supervision, Y.K. and L.M.; project administration, O.J.A., Y.K. and L.M.; funding acquisition, O.J.A.. All authors have read and agreed to the published version of the manuscript.

**Conflict of Interest:** The authors declare no conflict of interest regarding the research undertaken in this study.

**Data availability statement:** The data supporting this study's findings are available upon request. These data are managed within the Bank of Tanzania Supervision Information System (BSIS) and can be shared upon a formal request.

Source(s) of support/funding (if any): The authors received no direct funding for this study.

#### **Appendices** A

Dimension	FinTech Applications						
Payment	Mobile Payments	Third-Party Payments	QR Code Payments	Remittances	POS		
Resource allocation	Internet Loans	Network Investments	Online Lending's	P2PLoans			
Risk management	Internet Insurance	Online Financing	Regtech and suptech	e-KYC	Digital ID		
Network channel	Mobile Banking	Online Banking Service	Agency Banking	Internet Banking			
Big data	Bigdata	Data Mining	Bigdata analysis	Big Data Application			
Artificial intelligence	Intelligent Robot	Artificial Intelligence	Machine Learning				
Distributed technology	Cloud Computing	Block Chain Technology					

Appendix 1. FinTech Variable Dimensions and Applications.

Internet Technology	Vehicle Interconnection	Wireless	Mobile Internet	Mobile Communication
Security Technology	Biometrics	Fingerprint Identification		
	Note. As adopted fr	om: (Wang <i>et al.,</i> 2023; Wu <i>et d</i>	al., 2023; Lv et al., 2022	2; Wang et al., 2021; Gai et al.,
	2018).			

### References

- Abba, G. O., Okwa, E., Soje, B., & Aikpitanyi, L. N. (2018). Determinants of capital adequacy ratio of deposit money banks in Nigeria. *Journal of Accounting & Marketing*, 7(2), 1-7.
- Abdu, E. (2022). Financial distress situation of financial sectors in Ethiopia: A review paper. Cogent Economics & Finance, 10(1), 1996020. https://doi.org/10.1080/23322039.2021.1996020
- Abedifar, P., Molyneux, P., & Tarazi, A. (2018). Non-interest income and bank lending. *Journal of Banking & Finance*, 87, 411-426. https://doi.org/10.1016/j.jbankfin.2017.11.003
- Aboojafari, R., & Dehghani, M. (2022). Fintech approach to assessing innovative SMEs' credit risk. In *Innovative Finance for Technological Progress* (pp. 311-328). Routledge. https://doi.org/10.4324/9781003220220-18
- Adewumi, A., Ewim, S. E., Sam-Bulya, N. J., & Ajani, O. B. (2024). Advancing business performance through data-driven process automation: A case study of digital transformation in the banking sector. *International Journal of Multidisciplinary Research Updates*, 8(02). https://doi.org/10.53430/ijmru.2024.8.2.0049
- Aggarwal, M., Nayak, K. M., & Bhatt, V. (2023). Examining the factors influencing fintech adoption behavior of Gen Y in India. *Cogent Economics & Finance*, 11(1), 2197699. https://doi.org/10.1080/23322039.2023.2197699
- Alhassan, A. L., & Tetteh, M. L. (2017). Non-interest income and bank efficiency in Ghana: A two-stage DEA bootstrapping approach. *Journal of African Business*, 18(1), 124-142. https://doi.org/10.1080/15228916.2016.1227668
- Al-Shari, H. A., & Lokhande, M. A. (2023). The relationship between the risks of adopting FinTech in banks and their impact on the performance. *Cogent Business & Management*, 10(1), 2174242. https://doi.org/10.1080/23311975.2023.2174242
- An, T.-H. T., Chen, S.-H., & Yeh, K.-C. (2023). Does financial development enhance the growth effect of FDI? A multidimensional analysis in emerging and developing Asia. *International Journal of Emerging Markets*. https://doi.org/10.1108/IJOEM-03-2022-0495
- Anita, R., Wu, W., & Abdillah, M. R. (2023). Developing a Scale of Ethical Responsibility (SER): A Multi-Dimensional Instrument for Fintech Professionals. Social Indicators Research, 170(3), 1007-1033. https://doi.org/10.1007/s11205-023-03231-5
- Arellano, M., & Bover, O. (1995). Another look at the instrumental variable estimation of error-components models. Journal of Econometrics, 68(1), 29-51. https://doi.org/10.1016/0304-4076(94)01642-D
- Arnaut, D., & Bećirović, D. (2023). FinTech innovations as a disruptor of the traditional financial industry. In Digital Transformation of the Financial Industry: Approaches and Applications (pp. 233-254). Springer. https://doi.org/10.1007/978-3-031-23269-5\_14
- Asongu, S. A., Biekpe, N., & Cassimon, D. (2021). On the diffusion of mobile phone innovations for financial inclusion. *Technology in Society*, 65, 101542. https://doi.org/10.1016/j.techsoc.2021.101542
- Badunenko, O., Kumbhakar, S. C., & Lozano-Vivas, A. (2021). Achieving a sustainable cost-efficient business model in banking: The case of European commercial banks. *European Journal of Operational Research*, 293(2), 773-785. https://doi.org/10.1016/j.ejor.2020.12.039
- Banya, R., & Biekpe, N. (2018). Banking Efficiency and its determinants in selected frontier African markets. Economic Change and Restructuring, 51, 69-95. https://doi.org/10.1007/s10644-016-9200-3
- Blundell, R., & Bond, S. (1998). Initial conditions and moment restrictions in dynamic panel data models. *Journal of Econometrics*, 87(1), 115-143. https://doi.org/10.1016/S0304-4076(98)00009-8
- Boot, A., Hoffmann, P., Laeven, L., & Ratnovski, L. (2021). Fintech: What's old, what's new? *Journal of Financial Stability*, 53, 100836. https://doi.org/10.1016/j.jfs.2020.100836
- BOT. (2019). Bank of Tanzania Annual Report 2018/19. 2-10.
- BOT. (2020). Bank of Tanzania. Annual Report 2020/21, 7(8.5.2017), 2003-2005.
- Bu, Y., Du, X., Li, H., Yu, X., & Wang, Y. (2023). Research on the FinTech risk early warning based on the MS-VAR model: An empirical analysis in China. *Global Finance Journal*, 58, 100898. https://doi.org/10.1016/j.gfj.2023.100898
- Bunje, M. Y., Abendin, S., & Wang, Y. (2022). The multidimensional effect of financial development on trade in Africa: The role of the digital economy. *Telecommunications Policy*, 46(10), 102444. https://doi.org/10.1016/j.telpol.2022.102444
- Cao, Q., Zhu, T., & Yu, W. (2024). ESG investment and bank efficiency: Evidence from China. *Energy Economics*, 133, 107516. https://doi.org/10.1016/j.eneco.2024.107516
- Chan, R., Troshani, I., Hill, S. R., & Hoffmann, A. (2022). Towards an understanding of consumers' FinTech adoption: the case of Open Banking. *International Journal of Bank Marketing*. https://doi.org/10.1108/IJBM-08-2021-0397
- Chen, X., You, X., & Chang, V. (2021). FinTech and commercial banks' performance in China: A leap forward or survival of the fittest? *Technological Forecasting and Social Change*, 166(December 2020), 120645. https://doi.org/10.1016/j.techfore.2021.120645
- Cheng, M., & Qu, Y. (2020). Does bank FinTech reduce credit risk? Evidence from China. *Pacific Basin Finance Journal*, 63, 101398. https://doi.org/10.1016/j.pacfin.2020.101398

- Cheng, R. J. (2020). UTAUT implementation of cryptocurrency-based Islamic financing instrument. International Journal of Academic Research in Business and Social Sciences, 10(9), 873-884. https://doi.org/10.6007/IJARBSS/v10-i9/7880
- Cheriye, A. B. (2020). Measuring the efficiency of Pan African commercial banks: an application of Malmquist DEA methods. *Sciences*, 10(8), 942-954. https://doi.org/10.6007/IJARBSS/v10-i8/7665
- Cho, T.-Y., & Chen, Y.-S. (2021). The impact of financial technology on China's banking industry: An application of the meta frontier cost Malmquist productivity index. *The North American Journal of Economics and Finance*, 57, 101414. https://doi.org/10.1016/j.najef.2021.101414
- Dao, B. (2020). Bank capital adequacy ratio and bank performance in Vietnam: A simultaneous equations framework. *Journal of Asian Finance, Economics and Business*, 7(6), 39-46. https://doi.org/10.13106/jafeb.2020.vol7.no6.039
- Das Gupta, A., Sarker, N., & Rifat Rahman, M. (2021). Relationship among cost of financial intermediation, risk, and efficiency: Empirical evidence from Bangladeshi commercial banks. *Cogent Economics & Finance*, 9(1), 1967575. https://doi.org/10.1080/23322039.2021.1967575
- Dasilas, A., & Karanović, G. (2023). The impact of FinTech firms on bank performance: evidence from the UK. *EuroMed Journal of Business*. https://doi.org/10.1108/EMJB-04-2023-0099
- Davis, F. D. (1989). Technology acceptance model: TAM. Al-Suqri, MN, Al-Aufi, AS: Information Seeking Behavior and Technology Adoption, 205, 219.
- Deng, L., Lv, Y., Liu, Y., & Zhao, Y. (2021). Impact of fintech on bank risk-taking: Evidence from China. Risks, 9(5), 99. https://doi.org/10.3390/risks9050099
- Dong, Y., & Qin, S. J. (2018). A novel dynamic PCA algorithm for dynamic data modeling and process monitoring. *Journal of Process Control*, 67, 1-11. https://doi.org/10.1016/j.jprocont.2017.05.002
- Durak, İ., Çise, S. N., & Yazıcı, S. (2024). Developing a financial technology (FinTech) adoption scale: A validity and reliability study. *Research in International Business and Finance*, 70, 102344. https://doi.org/10.1016/j.ribaf.2024.102344
- Dwivedi, P., Alabdooli, J. I., & Dwivedi, R. (2021). Role of FinTech Adoption for Competitiveness and Performance of the Bank: A Study of Banking Industry in UAE. *International Journal of Global Business and Competitiveness*, 16(2), 1-9. https://doi.org/10.1007/s42943-021-00033-9
- Eibinger, T., Deixelberger, B., & Manner, H. (2024). Panel data in environmental economics: Econometric issues and applications to IPAT models. *Journal of Environmental Economics and Management*, 125, 102941. https://doi.org/10.1016/j.jeem.2024.102941
- Elsaid, H. M. (2023). A review of literature directions regarding the impact of fintech firms on the banking industry. *Qualitative Research in Financial Markets*, 15(5), 693-711. https://doi.org/10.1108/QRFM-10-2020-0197
- Ferreira, C. (2020). Evaluating European bank efficiency using data envelopment analysis: Evidence in the aftermath of the recent financial crisis. *International Advances in Economic Research*, 26(4), 391-405. https://doi.org/10.1007/s11294-020-09807-y
- Fianto, B. A., Hendratmi, A., & Aziz, P. F. (2021). Factors determining behavioral intentions to use Islamic financial technology: Three competing models. *Journal of Islamic Marketing*, 12(4), 794-812. https://doi.org/10.1108/JIMA-12-2019-0252
- Gai, K., Qiu, M., & Sun, X. (2018). A survey on FinTech. Journal of Network and Computer Applications, 103, 262-273. https://doi.org/10.1016/j.jnca.2017.10.011
- George, N. M., Ge, H., & Tang, D. (2024). Impact of digital financial inclusion on rural revitalization in Tanzania. Journal of Economics, Management and Trade, 30(6), 88-106. https://doi.org/10.9734/jemt/2024/v30i61216
- Gökgöz, F., Yalçın, E., & Salahaldeen, N. A. (2024). Investigating the financial efficiencies and productivities of the banking sector. Journal of Economic Studies, 51(5), 1036-1057. https://doi.org/10.1108/JES-07-2023-0338
- Gorton, G., & Winton, A. (2003). Financial intermediation. In *Handbook of the Economics of Finance* (Vol. 1, pp. 431-552). Elsevier. https://doi.org/10.1016/S1574-0102(03)01012-4
- Habib, A., Costa, M. D', Huang, H. J., Bhuiyan, M. B. U., & Sun, L. (2020). Determinants and consequences of financial distress: a review of the empirical literature. Accounting & Finance, 60, 1023-1075. https://doi.org/10.1111/acfi.12400
- Hsieh, M., & Lee, C. (2020). Bank liquidity creation, regulations, and credit risk. *Asia Pacific Journal of Financial Studies*, 49(3), 368-409. https://doi.org/10.1111/ajfs.12295
- Isiaku, L., & Adalier, A. (2024). Determinants of business intelligence systems adoption in Nigerian banks: The role of perceived usefulness and ease of use. *Information Development*, 02666669241307024. https://doi.org/10.1177/02666669241307024
- Isik, I., & Uygur, O. (2021). Financial crises, bank efficiency and survival: Theory, literature and emerging market evidence. International Review of Economics & Finance, 76, 952-987. https://doi.org/10.1016/j.iref.2021.07.016
- Kartikasary, M., Marsintauli, F., Serlawati, E., & Laurens, S. (2020). Factors affecting the non-performing loans in Indonesia. *Accounting*, 6(2), 97-106. https://doi.org/10.5267/j.ac.2019.12.003
- Kashif, M., Pinglu, C., Ullah, S., & Zaman, M. (2024). Evaluating the influence of financial technology (FinTech) on sustainable finance: a comprehensive global analysis. *Financial Markets and Portfolio Management*, 38(1), 123-155. https://doi.org/10.1007/s11408-023-00439-w
- Khalid, U., & Shafiullah, M. (2021). Financial development and governance: A panel data analysis incorporating cross-sectional dependence. *Economic Systems*, 45(2), 100855. https://doi.org/10.1016/j.ecosys.2021.100855
- Khan, H. H., Khan, S., & Ghafoor, A. (2023). Fintech adoption, the regulatory environment and bank stability: An empirical investigation from GCC economies. *Borsa Istanbul Review*, 23(6), 1263-1281. https://doi.org/10.1016/j.bir.2023.10.010
- Khan, H. H., Kutan, A. M., & Qureshi, F. (2024). Fintech Integration: Driving Efficiency in Banking Institutions across the Developing Nations. *Finance Research Letters*, 105772. https://doi.org/10.1016/j.frl.2024.105772

- Khan, S., Ullah, A., Pinglu, C., & Kashif, M. (2024). Impact of financial technology innovation on sustainable transition finance: the moderating role of globalization in BRI, advanced and emerging economies. *Clean Technologies and Environmental Policy*, 1-25. https://doi.org/10.1007/s10098-024-02898-9
- Kharrat, H., Trichilli, Y., & Abbes, B. (2024). Relationship between FinTech index and bank's performance: a comparative study between Islamic and conventional banks in the MENA region. *Journal of Islamic Accounting and Business Research*, 15(1), 172-195. https://doi.org/10.1108/JIABR-03-2022-0081
- Klus, M. F., Lohwasser, T. S., Holotiuk, F., & Moormann, J. (2019). Strategic alliances between banks and fintechs for digital innovation: Motives to collaborate and types of interaction. *The Journal of Entrepreneurial Finance* (JEF), 21(1), 1-23. https://doi.org/10.57229/2373-1761.1346
- Koloseni, D. N. (2021). User perceptions of mobile banking apps in Tanzania: Impact of information systems (IS) factors and customer personality traits. *The African Journal of Information and Communication*, 28, 1-26. https://doi.org/10.23962/10539/32214
- Koloseni, D., & Mandari, H. (2024). Expediting financial inclusion in Tanzania using FinTech: the diffusion of innovation theory perspective. *Technological Sustainability*, 3(2), 171-194. https://doi.org/10.1108/TECHS-11-2023-0048
- Kshetri, N. (2021). The role of artificial intelligence in promoting financial inclusion in developing countries. In *Journal of Global Information Technology Management* (Vol. 24, Issue 1, pp. 1-6). Taylor & Francis. https://doi.org/10.1080/1097198X.2021.1871273
- Ky, S. S., Rugemintwari, C., & Sauviat, A. (2019). Is fintech good for bank performance? The case of mobile money in the East African Community. SSRN Working Paper 3401930. https://doi.org/10.2139/ssrn.3401930
- La Torre, M. (2020). *Risk* in Banking: Developing a Knowledge Risk Management Framework for Cooperative Credit Banks. Springer Nature. https://doi.org/10.1007/978-3-030-54498-0
- Lasisi, A., & Attoh-Okine, N. (2018). Principal components analysis and track quality index: A machine learning approach. *Transportation Research Part C: Emerging Technologies*, 91, 230-248. https://doi.org/10.1016/j.trc.2018.04.001
- Latreche, H., Bellahcene, M., & Dutot, V. (2024). Does IT culture archetypes affect the usefulness and ease of use of e-banking services? A multistage approach of Algerian customers. *International Journal of Bank Marketing*. https://doi.org/10.1108/IJBM-02-2023-0100
- Lee, Chi-Chuan, Li, X., Yu, C.-H., & Zhao, J. (2021). Does fintech innovation improve bank efficiency? Evidence from China's banking industry. *International Review of Economics & Finance*, 74, 468-483. https://doi.org/10.1016/j.iref.2021.03.009
- Lee, Chien-Chiang, Ni, W., & Zhang, X. (2023). FinTech development and commercial bank efficiency in China. *Global Finance Journal*, 57, 100850. https://doi.org/10.1016/j.gfj.2023.100850
- Li, J., Li, H., Shang, X., Pu, Y., Zhang, M., & Pan, X. (2024). Exploring the impact of fintech, human capital, mineral policy, and institutional quality on poverty traps: Evidence from resource-dependent economies. *Resources Policy*, 93, 105015. https://doi.org/10.1016/j.resourpol.2024.105015
- Li, R., Li, L., & Zou, P. (2021). Credit risk shocks and banking efficiency: a study based on a bootstrap-DEA model with nonperforming loans as bad output. *Journal of Economic Studies*, 48(1), 1-19. https://doi.org/10.1108/JES-08-2019-0395
- Liao, C.-S. (2023). How does fintech affect bank efficiency in Taiwan? *Plos One*, 18(8), e0289629. https://doi.org/10.1371/journal.pone.0289629
- Lotto, J. (2018). The empirical analysis of the impact of bank capital regulations on operating efficiency. *International Journal of Financial Studies*, 6(2), 34. https://doi.org/10.3390/ijfs6020034
- Lotto, J. (2019). Evaluation of factors influencing bank operating efficiency in Tanzanian banking sector. *Cogent Economics & Finance*, 7(1), 1664192. https://doi.org/10.1080/23322039.2019.1664192
- Lv, S., Du, Y., & Liu, Y. (2022). How do fintechs impact banks' profitability?-an empirical study based on banks in China. *FinTech*, 1(2), 155-163. https://doi.org/10.3390/fintech1020012
- Ma, Y., & Lv, L. (2023). Financial development, financial instability, and fiscal policy volatility: International evidence. *The North American Journal of Economics and Finance*, 64, 101873. https://doi.org/10.1016/j.najef.2022.101873
- Makina, D. (2019). The potential of FinTech in enabling financial inclusion. In Extending financial inclusion in Africa (pp. 299-318). Elsevier. https://doi.org/10.1016/B978-0-12-814164-9.00014-1
- Maradin, D., Suljić Nikolaj, S., & Olgić Draženović, B. (2021). Efficiency and productivity of Islamic banking industry by using DEA method: a literature review. *The Changing Financial Landscape: Financial Performance Analysis of Real and Banking Sectors in Europe*, 205-217. https://doi.org/10.1007/978-3-030-82778-6\_12
- Medhioub, N., & Boujelbene, Y. (2024). Does digitalization moderate the link between innovation and economic growth? A two-step difference GMM analysis of developed and developing countries. *Journal of the Knowledge Economy*, 1-27. https://doi.org/10.1007/s13132-024-01963-4
- Msomi, T. S. (2022). Factors affecting non-performing loans in commercial banks of selected West African countries. *Banks and Bank Systems*, 17(1), 1. https://doi.org/10.21511/bbs.17(1).2022.01
- Muda, R., Lateff, M. S. M., Arshad, R., Rashdan, A. A., Oladapo, I. A., & Othman, J. (2021). Does Fintech revolution lead to the disintermediation of banks? A study into Islamic Bank income. *Fintech*, Digital Currency and the Future of Islamic Finance: Strategic, Regulatory and Adoption Issues in the Gulf Cooperation Council, 169-185. https://doi.org/10.1007/978-3-030-49248-9\_9
- Muganyi, T., Yan, L., Yin, Y., Sun, H., Gong, X., & Taghizadeh-Hesary, F. (2022). Fintech, regtech, and financial development: evidence from China. *Financial Innovation*, 8(1), 1-20. https://doi.org/10.1186/s40854-021-00313-6

- Murinde, V., Rizopoulos, E., & Zachariadis, M. (2022). The impact of the FinTech revolution on the future of banking: Opportunities and risks. *International Review of Financial Analysis*, 81, 102103. https://doi.org/10.1016/j.irfa.2022.102103
- Ntwiga, D. B. (2020). Fintech and banks collaboration: Does it influence efficiency in the banking sector? KBA Centre for Research on Financial Markets and Policy Working Paper Series.
- Odoom, R., & Kosiba, J. P. (2020). Mobile money usage and continuance intention among micro-enterprises in an emerging market - the mediating role of agent credibility. *Journal of Systems and Information Technology*, 22(1), 97-117. https://doi.org/10.1108/JSIT-03-2019-0062
- Ogbonna, A. E., Adediran, I. A., Oloko, T. F., & Isah, K. O. (2023). Information and Communication Technology (ICT) and youth unemployment in Africa. *Quality & Quantity*, 57(6), 5055-5077. https://doi.org/10.1007/s11135-022-01600-9
- Peres, F. A. P., & Fogliatto, F. S. (2018). Variable selection methods in multivariate statistical process control: A systematic literature review. Computers & Industrial Engineering, 115, 603-619. https://doi.org/10.1016/j.cie.2017.12.006
- Phan, D. H. B., Narayan, P. K., Rahman, R. E., & Hutabarat, A. R. (2020). Do financial technology firms influence bank performance? *Pacific-Basin Finance Journal*, 62, 101210. https://doi.org/10.1016/j.pacfin.2019.101210
- Prakash, N., Singh, S., & Sharma, S. (2021). Technological diffusion, banking efficiency and Solow's paradox: A frontier-based parametric and non-parametric analysis. *Structural Change and Economic Dynamics*, 58, 534-551. https://doi.org/10.1016/j.strueco.2021.07.007
- Rahman, M. M. (2024). Moderating effects of energy poverty on financial inclusion, FinTech lending, and economic growth: Evidence from fsQCA, NCA, and econometric models. *Environmental Challenges*, 15, 100867.
- Raj, K., Fredrick, D. P., Kurahattidesai, C., & Hegde, C. S. (2024). Artificial Intelligence Driven Customer Relationship Management: Harnessing the power of technology to improve business efficiency. *International Journal of Communication Networks and Information Security*, 16(4), 58-65.
- Rana, F., Hussain, B., Nassani, A. A., Hussain, A., Haffar, M., & Naqvi, S. A. A. (2024). Contribution of hydro energy production, economic complexity and technological innovation in achieving an environmentally sustainable *Asia. Geoscience Frontiers*, 15(5), 101876. https://doi.org/10.1016/j.gsf.2024.101876
- Rizvi, S. K. A., Rahat, B., Naqvi, B., & Umar, M. (2024). Revolutionizing finance: The synergy of fintech, digital adoption, and innovation. *Technological Forecasting and Social Change*, 200, 123112. https://doi.org/10.1016/j.techfore.2023.123112
- Rogers, E. M., Singhal, A., & Quinlan, M. M. (2014). Diffusion of innovations. In An Integrated Approach To Communication Theory and Research (pp. 432-448). Routledge.
- Roodman, D. (2009). How to do xtabond2: An introduction to difference and system GMM in Stata. *The Stata Journal*, 9(1), 86-136. https://doi.org/10.1177/1536867X0900900106
- Saâdaoui, F., & Khalfi, M. (2024). Revisiting Islamic banking efficiency using multivariate adaptive regression splines. *Annals of Operations Research*, 334(1), 287-315. https://doi.org/10.1007/s10479-022-04545-2
- Sajid, R., Ayub, H., Malik, B. F., & Ellahi, A. (2023). The Role of Fintech on Bank Risk-Taking: Mediating Role of Bank's Operating Efficiency. *Human Behavior and Emerging Technologies*, 2023(1), 7059307. https://doi.org/10.1155/2023/7059307
- Schreiber, J. B. (2021). Issues and recommendations for exploratory factor analysis and principal component analysis. *Research in Social and Administrative Pharmacy*, 17(5), 1004-1011. https://doi.org/10.1016/j.sapharm.2020.07.027
- Shah, W. U. H., Hao, G., Zhu, N., Yasmeen, R., Padda, I. U. H., & Abdul Kamal, M. (2022). A cross-country efficiency and productivity evaluation of commercial banks in South Asia: A meta-frontier and Malmquist productivity index approach. *Plos* One, 17(4), e0265349. https://doi.org/10.1371/journal.pone.0265349
- Shahbaz, M., İlarslan, K., & Yıldız, M. (2023). An econometric analysis of the relationship between financial development and carbon neutrality in Eurasian countries. In *Recent Developments in Green Finance, Green Growth and Carbon Neutrality* (pp. 129-148). Elsevier. https://doi.org/10.1016/B978-0-443-15936-7.00004-9
- Sheraz, M., Deyi, X., Mumtaz, M. Z., & Ullah, A. (2022). Exploring the dynamic relationship between financial development, renewable energy, and carbon emissions: A new evidence from belt and road countries. *Environmental Science and Pollution Research*, 29(10), 14930-14947. https://doi.org/10.1007/s11356-021-16641-0
- Siddique, A., Khan, M. A., & Khan, Z. (2022). The effect of credit risk management and bank-specific factors on the financial performance of the South Asian commercial banks. *Asian Journal of Accounting Research*, 7(2), 182-194. https://doi.org/10.1108/AJAR-08-2020-0071
- Sinaga, J. S., Muda, I., & Silalahi, A. S. (2020). The effect of BI rate, exchange rate, inflation, and third party fund (DPK) on credit distribution and its impact on nonperforming loan (NPL) on XYZ commercial segment bank. *Universal Journal of Accounting and Finance*, 8(3), 55-64. https://doi.org/10.13189/ujaf.2020.080301
- Singh, S. K., Basuki, B., & Setiawan, R. (2021). The effect of non-performing loan on profitability: Empirical evidence from Nepalese commercial banks. *The Journal of Asian Finance, Economics and Business*, 8(4), 709-716.
- Singh, S., Sahni, M. M., & Kovid, R. K. (2020). What drives FinTech adoption? A multi-method evaluation using an adapted technology acceptance model. *Management Decision*. https://doi.org/10.1108/MD-09-2019-1318
- Soud, N. S., & Aypek, N. (2020). Impact of corporate governance on banks efficiency using value added approach: The case from Tanzania. International Journal of Economics, *Commerce and Management*, 8(12), 1-13.
- Stulz, R. M. (2019). Fintech, bigtech, and the future of banks. *Journal of Applied Corporate Finance*, 31(4), 86-97. https://doi.org/10.1111/jacf.12378

- Sunardi, R., Hamidah, H., BUCHDADI, A. D., & Purwana, D. (2022). Factors determining adoption of fintech peer-to-peer lending platform: an empirical study in Indonesia. *Journal of Asian Finance, Economics and Business*, 9(1), 43-51.
- Temba, G. I., Kasoga, P. S., & Keregero, C. M. (2023). Corporate governance and financial performance: Evidence from commercial banks in Tanzania. *Cogent Economics & Finance*, 11(2), 2247162. https://doi.org/10.1080/23322039.2023.2247162
- Theiri, S., & Alareeni, B. (2023). Perception of the digital transformation as a strategic advantage through the Covid 19 crisis? case of Tunisian banks. *Journal of Sustainable Finance & Investment*, 13(1), 477-498. https://doi.org/10.1080/20430795.2021.1964809
- Van Phuc Le, S. H. Do, & Nguyen, H. N. L. (2022). A study on the factors affecting intention of using online banking services in vietnam. An Thinh Nguyen, 179.
- Wang, H., Mao, K., Wu, W., & Luo, H. (2023). Fintech inputs, non-performing loans risk reduction, and bank performance improvement. *International Review of Financial Analysis*, 90, 102849.
- Wang, Y., Xiuping, S., & Zhang, Q. (2021). Can fintech improve the efficiency of commercial banks?-An analysis based on big data. *Research in International Business and Finance*, 55, 101338. https://doi.org/10.1016/j.ribaf.2020.101338
- Wirtz, J., Hofmeister, J., Chew, P. Y. P., & Ding, X. (2023). Digital service technologies, service robots, AI, and the strategic pathways to cost-effective service excellence. *The Service Industries Journal*, 43(15-16), 1173-1196. https://doi.org/10.1080/02642069.2023.2226596
- Wu, X., Jin, T., Yang, K., & Qi, H. (2023). The impact of bank FinTech on commercial banks' risk-taking in China. International Review of Financial Analysis, 90, 102944. https://doi.org/10.1016/j.irfa.2023.102944
- Yao, T., & Song, L. (2023). Fintech and the economic capital of Chinese commercial bank's risk: Based on theory and evidence. International Journal of Finance & Economics, 28(2), 2109-2123. https://doi.org/10.1002/ijfe.2528
- Yitayaw, M. K. (2021). Firm-specific and macroeconomic determinants of commercial banks liquidity in Ethiopia: Panel data approach. Cogent Business & Management, 8(1), 1956065. https://doi.org/10.1080/23311975.2021.1956065
- Zenebe Lema, T. (2017). Determinants of bank technical efficiency: Evidence from commercial banks in Ethiopia. Cogent Business & Management, 4(1), 1268356. https://doi.org/10.1080/23311975.2016.1268356
- Zhao, J., Li, X., Yu, C.-H., Chen, S., & Lee, C.-C. (2022). Riding the FinTech innovation wave: FinTech, patents and bank performance. *Journal of International Money and Finance*, 122, 102552. https://doi.org/10.1016/j.jimonfin.2021.102552
- Zhao, Y., Goodell, J. W., Wang, Y., & Abedin, M. Z. (2023). Fintech, macroprudential policies and bank risk: Evidence from China. International Review of Financial Analysis, 87, 102648. https://doi.org/10.1016/j.irfa.2023.102648
- Zheng, C., Rahman, M. A., Hossain, S., & Alam Siddik, M. N. (2024). Construction of a composite fintech index to measure financial inclusion for developing countries. *Applied Economics*, 1-18. https://doi.org/10.1080/00036846.2024.2313600
- Zhu, N., Shah, W. U. H., Kamal, M. A., & Yasmeen, R. (2021). Efficiency and productivity analysis of Pakistan's banking industry: A DEA approach. International *Journal of Finance & Economics*, 26(4), 6362-6374. https://doi.org/10.1002/ijfe.2123
- Zuo, L., Li, H., & Xia, X. (2023). An empirical analysis of the impact of digital finance on the efficiency of commercial banks. *Sustainability*, 15(5), 4667. https://doi.org/10.3390/su15054667

**Disclaimer:** All statements, viewpoints, and data featured in the publications are exclusively those of the individual author(s) and contributor(s), not of MFI and/or its editor(s). MFI and/or the editor(s) absolve themselves of any liability for harm to individuals or property that might arise from any concepts, methods, instructions, or products mentioned in the content.